Author’s response to reviews

Title: Depressive symptoms, Atherosclerotic burden and Cerebral Blood Flow disturbances in a cohort of Octogenarian men from a general population.

Authors:

Arkadiusz Siennicki-Lantz (Arkadiusz.Siennicki-Lantz@med.lu.se)
Lena André-Petersson (Lena.Andre-Petersson@med.lu.se)
Per Wollmer (Per.Wollmer@med.lu.se)
Sôleve Elmståhl (Solve.Elmstahl@med.lu.se)

Version: 4 Date: 16 December 2013

Author’s response to reviews:

According to editorial comments we completes our manuscript with a chapter on ethics (2.6) with a name of ethical commite and consent of the study subjects.

According to the comment from one referee to be more cautious in the conclusions, considering the small number of depressed subjects and moreover with mild symptoms: we reformulated slightly Conclusion-chapter at the end of the manuscript and one in the abstract